Author's response to reviews

Title: The Home-Based Older People's Exercise (HOPE) trial: study protocol for a randomised controlled trial

Authors:

Andrew Clegg (andrewpaulclegg@yahoo.co.uk)
Sally Barber (sally.barber@bradfordhospitals.nhs.uk)
John Young (john.young@bradfordhospitals.nhs.uk)
Anne Forster (a.forster@leeds.ac.uk)
Steve Iliffe (s.iliffe@ucl.ac.uk)

Version: 4 Date: 1 June 2011

Author's response to reviews: see over
Many thanks for your email confirming that our manuscript is potentially ready to be accepted. We have made the necessary modifications to ensure that our manuscript conforms to the journal style. Specifically, we have amended the protocol title, moved the list of abbreviations to the end of the manuscript and added the trial registration information at the end of the abstract.

If there are any further amendments required please do not hesitate to contact us.

Kind regards

Andrew Clegg, on behalf of the HOPE trial team